Author's response to reviews

Title: Breast and prostate cancer: an analysis of common epidemiological features in mortality trends in Spain

Authors:

Gonzalo López-Abente (glabente@isciii.es)
Sergio Mispireta (smispireta@isciii.es)
Marina Pollán (mpollan@isciii.es)

Version: 5 Date: 30 October 2014

Author's response to reviews:

EDITORS COMMENTS:

1. page 2, line 50, please change 'respectivelly' to respectively.

OK. Done.

2. "Please clarify who permitted your use of mortality data from the Spanish National Statistics Institute, in the methods section of your manuscript."

Spanish National Statistics Institute mortality data used in this article are public and are not under to any confidentiality clause.